\begin{thebibliography}{10}

\bibitem{Agarwal95}
A.~Agarwal, R.~Bianchini, D.~Chaiken, K.~L. Johnson, D.~Kranz, J.~Kubiatowicz,
  B.-H. Lim, K.~Mackenzie, , and D.~Yeung.
\newblock {The MIT Alewife Machine: Architecture and Performance}.
\newblock In {\em Proceedings of the 22nd Annual International Symposium on
  Computer Architecture}, pages 2 -- 13, 1995.

\bibitem{Agerwala95}
T.~Agerwala, J.~L. Martin, J.~H. Mirza, D.~C. Sadler, D.~M. Dias, and M.~Snir.
\newblock {{SP2} System Architecture}.
\newblock {\em {IBM} Systems Journal}, 34(2):152 -- 184, 1995.

\bibitem{Barton94}
E.~Barton, J.~Cownie, and M.~{McLaren}.
\newblock Message passing on the meiko cs-2.
\newblock {\em Parallel Computing}, 20:497 -- 507, 1994.

\bibitem{Blumrich94a}
M.~A. Blumrich, K.~Li, R.~Alpert, C.~Dubnicki, E.~W. Felten, and J.~Sandberg.
\newblock {Virtual Memory Mapped Network Interface for the SHRIMP
  Multicomputer}.
\newblock In {\em Proceedings of the 21st International Symposium on Computer
  Architecture}, pages 142 -- 153, Apr. 1994.

\bibitem{Boden95}
N.~J. Boden, D.~Cohen, R.~E. Felderman, A.~E. Kulawik, C.~L. Seitz, J.~N.
  Seizovic, and W.~Su.
\newblock {Myrinet -- A gigabit-per-Second Local-Area Network}.
\newblock {\em {{IEEE}} Micro}, Feb. 1995.

\bibitem{Borkar88}
S.~Borkar, R.~Cohn, G.~Cox, S.~Gleason, T.~Gross, H.~T. Kung, M.~Lam, B.~Moore,
  C.~Peterson, J.~Pieper, L.~Rankin, P.~S. Tseng, J.~Sutton, J.~Urbanski, , and
  J.~Webb.
\newblock {iWarp: An Integrated Solution to High-speed Parallel Computing}.
\newblock In {\em Proceedings of Supercomputing '88, Orlando, Florida}, pages
  330 -- 339, Nov. 1988.

\bibitem{Borkar90}
S.~Borkar, R.~Cohn, G.~Cox, T.~Gross, H.~T. Kung, M.~Lam, M.~Levine, B.~Moore,
  W.~Moore, C.~Peterson, J.~Susman, J.~Sutton, J.~Urbanski, and J.~Webb.
\newblock {Supporting Systolic and Memory Communication in iWarp}.
\newblock In {\em Proceedings of the 17th International Symposium on Computer
  Architecture}, pages 70 -- 81, May 1990.

\bibitem{Boughton94}
G.~A. Boughton.
\newblock {Arctic Routing Chip}.
\newblock In {\em Proceedings of Hot Interconnects II, Stanford, CA}, pages 164
  -- 173, Aug. 1994.

\bibitem{Brewer95}
T.~Brewer.
\newblock {A Highly Scalable System Utilizing up to 128 PA-RISC Processors}.
\newblock Technical report, Convex Computer Corporation, 1995.
\newblock (Available from http://www.convex.com/tech\_cache/technical.html in
  Jul 96.).

\bibitem{Chang96b}
C.-C. Chang, G.~Czajkowski, and T.~{von Eicken}.
\newblock {Design and Performance of Active Messages on the IBM SP-2}.
\newblock (Work at Dept of Computer Science, Cornell University, Ithaca.
  Available from http://www.cs.cornell.edu/Info/Projects/CAM/ in Jul 96).

\bibitem{Chiou95}
D.~Chiou, B.~S. Ang, R.~Greiner, Arvind, J.~C. Hoe, M.~J. Beckerle, J.~E.
  Hicks, and A.~Boughton.
\newblock {StarT-NG: Delivering Seamless Parallel Computing}.
\newblock In {\em Proceedings of the First International EURO-PAR Conference,
  Stockholm, Sweden}, pages 101 -- 116, Aug. 1995.

\bibitem{Exemplar93}
Convex Computer Corporation, Richardson, Tx.
\newblock {\em {Exemplar Architecture}}, Nov. 1993.

\bibitem{Culler93}
D.~Culler, R.~Karp, D.~Patterson, A.~Sahay, K.~E. Schauser, E.~Santos,
  R.~Subrmonian, and T.~von Eicken.
\newblock {{LogP}: Torwards a Realistic Model of Parallel Computation}.
\newblock In {\em Fourth {ACM} {SIGPLAN} Symposium on Principles and Practice
  of Parallel Programming, San Diego}, pages 1--12, 1993.

\bibitem{Culler96}
D.~Culler, L.~T. Liu, R.~Martin, and C.~Yoshikawa.
\newblock {LogP Performance Assessment of Fast Network Interfaces}.
\newblock {\em {IEEE} Micro}, 1996.
\newblock (to appear).

\bibitem{Culler90}
D.~E. Culler and G.~M. Papadopoulos.
\newblock The {Explicit Token Store}.
\newblock {\em Journal of Parallel and Distributed Computing}, 10(4):289--308,
  1990.

\bibitem{Dally92}
W.~J. Dally, R.~Davison, J.~A.~S. Fiske, G.~Fyler, J.~S. Keen, R.~A. Lethin,
  M.~Noakes, and P.~R. Nuth.
\newblock {The Message-Driven Processor: A Multicomputer Processing Node with
  Efficient Mechanisms}.
\newblock {\em {IEEE} Micro}, Apr. 1992.

\bibitem{Fillo95}
M.~Fillo, S.~W. Keckler, W.~J. Dally, N.~P. Carter, A.~Chang, Y.~Gurevich, and
  W.~S. Lee.
\newblock {The M-Machine Multicomputer}.
\newblock In {\em Proceedings of the 28th Annual International Symposium on
  Microarchitecture, Ann Arbor, Michigan}, 1995.

\bibitem{Gillett96}
R.~B. Gillett.
\newblock {Memory Channel Network for PCI}.
\newblock {\em {IEEE} Micro}, pages 12 -- 18, Feb. 1996.

\bibitem{Hoe94}
J.~C. Hoe.
\newblock {Network Interface for Message-Passing Parallel Computation on a
  Workstation Cluster}.
\newblock In {\em Proceedings of Hot Interconnects II, Stanford, CA}, pages 154
  -- 163, Aug. 1994.

\bibitem{Hoe96}
J.~C. Hoe and M.~Ehrlich.
\newblock {StarT-Jr: A Parallel System from Commodity Technology}.
\newblock {CSG Memo} 384, MIT Laboratory for Computer Science, July 1996.

\bibitem{ibm-sp1}
IBM.
\newblock {\em {IBM Scalable POWERparallel System Reference Guide}}, 1993.
\newblock IBM Publication Number G325-0648-00.

\bibitem{Paragon-91}
Intel Corporation, Intel Corporation, Supercomputer Systems Division, 15201
  N.W. Greenbrier Parkway, Beaverton, OR 97006.
\newblock {\em Paragon XP/S Product Overview}, 1991.

\bibitem{Joerg92}
C.~F. Joerg and D.~S. Henry.
\newblock {A Tightly-Coupled Processor-Network Interface}.
\newblock In {\em Proceedings of the Fifth International Conference on
  Architectural Support for Programming Languages and Operating Systems,
  Boston, MA}, pages 111--122, Oct. 1992.

\bibitem{Karamcheti}
V.~Karamcheti and A.~Chien.
\newblock {FM: Fast Messaging on the Cray T3D}.
\newblock (Work at Concurrent Systems Architecture Group, Dept of Computer
  Science, University of Illinois at Urbana-Champaign. Available from
  http://www-csag.cs.uiuc.edu/projects/comm/fm.html in Jul 96).

\bibitem{Kessler93}
R.~E. Kessler and J.~L. Schwarzmeier.
\newblock {Cray T3D: a New Dimension for Cray Research}.
\newblock In {\em Digest of Papers, {COMPCON} Spring 93, San Francisco, CA},
  pages 176 -- 182, Feb. 1993.

\bibitem{Kuskin94}
J.~Kuskin, D.~Ofelt, M.~Heinrich, J.~Heinlein, R.~Simoni, K.~Gharachorloo,
  J.~Chaplin, D.~Nakahira, J.~Baxter, M.~Horowitz, A.~Gupta, M.~Rosenblum, and
  J.~Hennessy.
\newblock {The {Stanford} {FLASH} Multiprocessor}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, Apr. 1994.

\bibitem{Leiserson85}
C.~E. Leiserson.
\newblock {Fat-trees: Universal Networks for Hardware-efficient
  Supercomputing}.
\newblock {\em {IEEE} Transactions on Computers}, C-34(10), Oct. 1985.

\bibitem{Leiserson92}
C.~E. Leiserson et~al.
\newblock {The Network Architecture of the Connection Machine {CM-5}}.
\newblock In {\em Proceedings of the 1992 ACM Symposium on Parallel Algorithms
  and Architectures}, 1992.

\bibitem{Lenoski92}
D.~Lenoski, J.~Laudon, T.~Joe, D.~Nakahira, L.~Stevens, A.~Gupta, and
  J.~Hennessy.
\newblock {The DASH Prototype: Implementation and Performance}.
\newblock In {\em Proceedings of the 19th Annual International Symposium on
  Computer Architecture, Gold Coast, Australia}, pages 92 -- 103, 1992.

\bibitem{Liu94}
L.~T. Liu and D.~E. Culler.
\newblock {Measurements of Active Messages Performance on the {CM-5}}.
\newblock (Work at Dept of Computer Science, University of California at
  Berkeley. Available from http://now.cs.berkeley.edu/Papers/papers.html in Jul
  96), May 1994.

\bibitem{Liu95}
L.~T. Liu and D.~E. Culler.
\newblock {Evaluation of the Intel Paragon on Active Message Communication}.
\newblock In {\em Proceedings of Intel Supercomputer Users Group Conference},
  June 1995.

\bibitem{Martin94}
R.~P. Martin.
\newblock {HPAM: An Active Message Layer for a Network of {HP} Workstations}.
\newblock In {\em Proceedings of Hot Interconnects II, Stanford, CA}, pages 40
  -- 58, Aug. 1994.

\bibitem{ppc-arch94}
C.~May, E.~Silha, R.~Simpson, and H.~Warren, editors.
\newblock {\em {The PowerPC Architecture: A Specification for a New Family of
  RISC Processors}}.
\newblock Morgan Kaufman Publishers, Inc., San Francisco, CA, second edition,
  May 1994.

\bibitem{Mukherjee96}
S.~S. Mukherjee, B.~Falsafi, M.~D. Hill, and D.~A. Wood.
\newblock {Coherent Network Interfaces for Fine-Grain Communication}.
\newblock In {\em Proceedings of the 23rd International Symposium on Computer
  Architecture}, May 1996.

\bibitem{Pakin95}
S.~Pakin, M.~Lauria, and A.~Chien.
\newblock {High Performance Messaging on Workstations: Illinois Fast Messages
  (FM) for Myrinet}.
\newblock In {\em Proceedings of Supercomputing '95, San Diego, CA}, 1995.

\bibitem{Papadopoulos93}
G.~M. Papadopoulos, G.~A. Boughton, R.~Greiner, and M.~J. Beckerle.
\newblock {{*T}: Integrated Building Blocks for Parallel Computing}.
\newblock In {\em Proceedings of Supercomputing '93, Portland, Oregon}, pages
  624--635, Nov. 1993.

\bibitem{Reinhardt94}
S.~K. Reinhardt, J.~R. Larus, and D.~A. Wood.
\newblock {Tempest and Typhoon: User-Level Shared Memory}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, pages 325 -- 336, Apr. 1994.

\bibitem{sakai89}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, Y.~Kodama, and T.~Yuba.
\newblock {An Architecture of a Dataflow Single Chip Processor}.
\newblock {\em 16th Annual International Symposium on Computer Architecture},
  pages 46--53, 1989.

\bibitem{Scott96}
S.~L. Scott.
\newblock {Synchronization and Communication in the T3E Multiprocessor}.
\newblock In {\em Proceedings of the Seventh International Conference on
  Architectural Support for Programming Languages and Operating Systems,
  Cambridge, MA}, pages 26 -- 36, Oct. 1996.

\bibitem{Numa-q}
Sequent Computer Systems, Inc.
\newblock {\em {Sequent's NUMA-Q Architecture White Paper}}.
\newblock (Available from
  http://www.sequent.com/public/solution/numaq/technology/archindex.html in Jul
  96, ).

\bibitem{Sobalvarro95}
P.~G. Sobalvarro and W.~E. Weihl.
\newblock {Demand-based Coscheduling of Parallel Jobs on Multiprogrammed
  Multiprocessors }.
\newblock In {\em Job Scheduling Strategies for Parallel Processing, Lecture
  Notes in Computer Science Vol. 949}. Springer-Verlag, 1995.

\bibitem{Stunkel95}
C.~B. Stunkel, D.~G. Shea, B.~Abali, M.~G. Atkins, C.~A. Bender, D.~G. Grice,
  P.~Hochschild, D.~J. Joseph, B.~J. Nathanson, R.~A. Swetz, R.~F. Stucke,
  M.~Tsao, and P.~R. Varker.
\newblock {The {SP2} High-Performance Switch}.
\newblock {\em {IBM} Systems Journal}, 34(2):185 -- 204, 1995.

\bibitem{cm-5}
Thinking Machines Corporation, 245 First Street, Cambridge, MA02142, USA.
\newblock {\em {Connection Machine {CM-5} Technical Summary}}, Nov. 1993.

\bibitem{vonEicken92}
T.~von Eicken, D.~E. Culler, S.~C. Goldstein, and K.~E. Schauser.
\newblock {Active Messages: a Mechanism for Integrated Communication and
  Computation}.
\newblock In {\em Proceedings of the 19th Annual International Symposium on
  Computer Architecture Conference Proceedings, Gold Coast, Australia}, pages
  256 -- 266, 1992.

\bibitem{Whitby-Strevens85}
C.~Whitby-Strevens.
\newblock {The Transputer}.
\newblock In {\em Proceedings of the 12th Annual International Symposium on
  Computer Architecture}, pages 292 -- 300, 1985.

\end{thebibliography}


%======================================================================
\section{Related Work}
\label{sec:rl-wk}
%======================================================================

The network interface unit in a parallel system can be located at one
of four places, each at a different distance from the processor: (i)
in the processor core; (ii) on a cache interface; (iii) on the
cache-coherent memory bus; or (iv) on the I/O bus.  

Though good for small to medium size messages, none of the on-chip (in
the processor core) network interfaces provide special bulk transfer
hardware.  Almost all these machines are proof-of-concept experimental
platforms where multi-tasking is rarely addressed.  Multi-threaded
sharing of message queue is not addressed except among the Dataflow
machines; these machines support only small messages sent and
(implicitly) received with individual instructions.  Examples of
processors with integrated network interfaces include
transputers\cite{Whitby-Strevens85}, the iWarp systolic
processor\cite{Borkar88,Borkar90}, dataflow processors like
Monsoon\cite{Culler90} and the EMC-R\cite{sakai89}, and hybrid
processors such as the MDP\cite{Dally92}, M-machine\cite{Fillo95}, and
the MIT/Motorola 88110MP\cite{Papadopoulos93}.

The Alewife\cite{Agarwal95} is an example of a network interface on
the L1 cache interface.  Alewife supports multi-part message
specification and its Sparcle processor provides hardware
multi-threading.  Race conditions during access to network interface
is prevented by disabling multi-threading during each message
interface access critical section.  
%Not much attention is paid to
%multi-tasking in Alewife.  
The design for StarT-NG\cite{Chiou95} proposed a network interface to
the L2 cache interface but only one message type is
supported.

Placing the network interfaces on I/O buses designed to accommodate
third party devices, is currently a popular approach.  Commercial
examples include machines like the IBM SP-1\cite{ibm-sp1} and
SP-2\cite{Agerwala95,Stunkel95}, and plug-in extensions like Myricom's
Myrinet\cite{Boden95} and DEC's Memory Channel\cite{Gillett96}, while
research efforts include Princeton's SHRIMP\cite{Blumrich94a} and
StarT-Jr\cite{Hoe94}.  Most I/O bus-based message passing support
utilizes main memory DRAM to house user-level message queues.  The NIU
delivers message between such queues on different nodes.  In some
designs, messages are sent out by writing directly across the I/O
bridge chip to the NIU.  Although these projects would like to provide
support for the three Ms, research efforts have so far been more
concerned about overcoming the challenge of traversing at least two
buses, and crossing one or more bridge chips before even reaching the
network interface.  With multiple user-level message queues housed in
main memory, most I/O bus based NIUs support multi-tasking fairly
well.  Multi-threaded access to the same message queue generally has
no special support.

%\comment{Say something about Utah Avalanche\cite{Swanson96}}

Our work is focused on network interface connected to the memory bus.
Many of the machines that support distributed shared memory locate
their network interface unit (NIU) on the memory bus.  Among them,
DASH\cite{Lenoski92}, Exemplar\cite{Exemplar93,Brewer95}, and
NUMA-Q\cite{Numa-q} do not directly support message passing.  Message
passing on the T3D\cite{Kessler93} is implemented with its remote
memory access and atomic fetch-and-add capabilities, the latter for
remote message queue space acquisition.  Message data is transfered
using the machine's shared memory capability.  Multi-granularity
support includes a bulk transfer engine, and special support for
barrier and Eureka synchronizations.  Its successor, the
T3E\cite{Scott96} provides more traditional form of message passing --
messages are launched from special off-chip E registers, and received
from buffers in main memory.  Support for multiple receive queues
facilitates multi-tasking in both the T3D and T3E.
FLASH\cite{Kuskin94} has a programmable memory controller, which can
be harnessed to implement a virtual memory mapped NIU.  Just about any
form of NIU can be implemented in firmware.  Thus far, their planned
message passing support appears to be the memory to memory bulk copy
form.  Typhoon\cite{Reinhardt94}, while having some hardware support
for message passing, is primarily a shared memory machine.

The {CM-5}\cite{cm-5} is a message passing machine with very
respectable performance for its time.  However, it does not handle
larger messages (messages bigger than a few words) efficiently, and
multi-tasking support, though present, is not flexible.  The
Paragon\cite{Paragon-91} uses a second processor at each node to copy
messages between the otherwise unprotected hardware network interface
and software message queues in main memory.  This provides protection
in a multi-tasking environment but introduces the overhead of
copying.
% -- a message can
%traverse the memory bus as many as three times on its way out of the
%sending node, and another three times on its way in.  It also requires
%polling by the coprocessor which introduces delays and extra bus
%occupancy.  
The Meiko CS-2\cite{Barton94} also has a message
coprocessor, and operates in a similar way as the Paragon, though its
design is more efficient in that the coprocessor connects directly to
the network.

More recent work by Mukherjee et. al.\cite{Mukherjee96} on Coherent
Network Interface (CNI), presents a nice and efficient network
interface.  The design however does not address the issue of sharing a
message queue among several concurrent threads.  It also does not
adapt the data transfer mechanism as communication granularity
changes.  Multi-tasking, though briefly mentioned, was not discussed
in detail.

Figure~\ref{tab: mp-machines-cp} provides a {\em very rough}
comparison of \start/'s message passing performance compared to other
machines.  Aside from \start/'s numbers, the numbers in
Figure~\ref{tab: mp-machines-cp} are reported by other researchers,
and were measured either with Active Message (AM)\cite{vonEicken92} or
Illinois Fast Messages (FM)\cite{Pakin95}, both of which are very fast
message passing libraries, and are taken from many
articles\cite{Liu94,Martin94,Liu95,Culler96,Chang96b,Hoe96,Karamcheti,Pakin95}.
It is impossible to conduct a completely fair comparison because the
machines are from different time periods, not all the measurements
reported are for the same message size (varies from zero to several
words of payload), and some implementations include sliding window
protocols to deal with lossy network hardware which also provides
flow-control as a side benefit.  Nevertheless, the table does provide
some insight into how \start/'s performance compares with these other
machines.

\begin{figure}
\begin{tabular}{|l|c|c|c|c|c|c|}
%
\hline
			        &	    &           & one-way        &	    &	 	&	\\
 				& processor & Interface & nearest        &  Send    &  Receive 	&  Peak	\\
\multicolumn{1}{|c|}{Machine}   &   speed   &  Location & neighbor       & Overhead & Overhead 	&  Bandwidth \\
			        &  (MHz)    &           & latency 	 & ($\mu$s) & ($\mu$s)	& (MByte/sec)\\
			        &	    &           & ($\mu$s)       &	    &	 	&	\\
\hline
%
\hline
{\it \start/ (AM)}		
&{\it 150} & {\it memory bus} & {\it 1.25} & {\it 0.13} & {\it 0.51} & {\it 140}\\
\hline
{CM-5} (CMAM)\cite{Liu94}     
& 33 & memory bus &  5.7  &   1.5    &   1.25   &    10  \\
\hline
Paragon (AM)\cite{Liu95} 	
& 50 & memory bus & 8.4   &   2.2    &   1.0    &   145  \\
\hline
Meiko CS-2 (AM)\cite{Culler96}
& 66 & memory bus & 10.8  &   1.7    &   1.6    &    39  \\
\hline
T3D (FM)\cite{Karamcheti}        
& 200 & memory bus &  10   & \multicolumn{2}{|c|}{4} & \NA \\
\hline
%----------
\hline
HP+Medusa (AM)\cite{Martin94}
& 99 & graphics bus & 10.15 & \NA     &   \NA     &    12  \\
\hline
%----------
\hline
StarT-Jr (AM)\cite{Hoe96}   
&120 & I/O bus    &  27   &   1.5    &   1.1    &    7 \\ 
\hline
Sparc20+ATM (AM)\cite{Chang96b}
& 50 & I/O bus    &  33   & \multicolumn{2}{|c|}{3} & 14  \\
\hline
SP-2 (AM)\cite{Chang96b}            
& 66 & I/O bus    & 25.5  & \multicolumn{2}{|c|}{4} & 34 \\
\hline
Sparc20+Myrinet(AM)\cite{Culler96} 
& 50 & I/O bus & 15.7 &   2.0    &   2.6    &    20 \\
\hline
Sparc20+Myrinet(FM)\cite{Pakin95} 
& 50 & I/O bus & 25   &    \NA    &   \NA     &    20 \\
\hline
%(?)Myrinet2 (FM)13.1 us(n)      <-- 4.4us -->                            17.5MB/sec
%\hline
%
\end{tabular}
\caption{ Message passing performance of several representative
systems.  \start/'s numbers are estimates obtained in the fashion
described in Section~\protect{\ref{sec:performance}}. }
\label{tab: mp-machines-cp}
\end{figure}

%======================================================================
%                                 END
%======================================================================